Title: What is the value and impact of quality and safety teams? A scoping review.

Authors:

Deborah E White (dwhit@ucalgary.ca)
Sharon E Straus (sharon.straus@utoronto.ca)
H Tom Stelfox (Tom.Stelfox@albertahealthservices.ca)
Jayna M Holroyd-Leduc (Jayna.Holroyd-Leduc@albertahealthservices.ca)
Chaim M Bell (Bellc@smh.toronto.on.ca)
Karen Jackson (karen.jackson@albertahealthservices.ca)
Jill M Norris (jmorris@ucalgary.ca)
W Ward Flemons (flemons@ucalgary.ca)
Michael E Moffatt (MMoffatt@wrha.mb.ca)
Alan J Forster (aforster@ohri.ca)

Version: 9 Date: 18 August 2011

Author's response to reviews:

Page 17, line 11 - now reads "contributions from Faculty of Nursing, University of Calgary, Winnipeg Regional Health "

Page 23, line 7 - now reads "100. Hobde BL, Hoffman PB, Makens PK, Tecca MB: Pursuing clinical and operational"